Author's response to reviews

Title: Regulation of sonic hedgehog-GLI1 downstream target genes PTCH1, Cyclin D2, Plakoglobin, PAX6 and NKX2.2 and their epigenetic status in medulloblastoma and astrocytoma.

Authors:

Mehdi H Shahi (mehdihayat@gmail.com)
Mohammad Afzal (mhmmd.afzal@gmail.com)
Subrata Sinha (sub.sinha@hotmail.com)
Charles G Eberhart (ceberha@jhmi.edu)
Juan A Rey (jarey.hulp@salud.madrid.org)
Xing Fan (xingf@med.umich.edu)
Javier S Castresana (jscastresana@unav.es)

Version: 11 Date: 2 November 2010

Author's response to reviews:

Please, receive the final copy, once modified according to the suggestions received by email from the editorial office.

Sincerely,

Javier S Castresana